Reconsidering the relationship between Gulf Stream transport and dynamic sea level at U.S. East Coast

Lequan Chi¹, Christopher Lee Pitt Wolfe², and Sultan Hameed³

¹University of Georgia ²School of Marine and Atmospheric Sciences ³Inst. for Terrestrial and Planetary Atmospheres

November 23, 2022

Abstract

The relationship between Gulf Stream (GS) transport and coastal sea level is investigated using monthly GS transport between 1993–2019 at Florida Straits and ten altimeter tracks. The results show that GS transport decorrelates quickly along its path, indicating it is misleading to assume that transport at a particular location represents strength of the GS as a whole. GS transport south of Cape Hatteras is significantly correlated with coastal sea level in both the South Atlantic Bight (SAB) and Middle Atlantic Bight (MAB). However, the significant correlations in MAB are due to their concurrent response wind on the shelf—the correlation becomes insignificant once the influence of local winds is removed. North of Cape Hatteras, the influence of GS transport on sea level is mostly over the deep ocean and rarely on the shelf, indicating that there is no dynamic link between the GS transport and coastal sea level in this region.

Hosted file

essoar.10508225.1.docx available at https://authorea.com/users/538042/articles/601990-reconsidering-the-relationship-between-gulf-stream-transport-and-dynamic-sea-level-at-u-s-east-coast

Hosted file

manuscipt_sealevel_gs_v1.16c2_supporting_information.docx available at https://authorea. com/users/538042/articles/601990-reconsidering-the-relationship-between-gulf-streamtransport-and-dynamic-sea-level-at-u-s-east-coast

Reconsidering the relationship between Gulf Stream transport and dynamic sea level at U.S. East Coast

Lequan Chi¹, Christopher L. P. Wolfe², and Sultan Hameed²

¹Skidaway Institute of Oceanography, University of Georgia, Savannah, GA 31411

²School of Marine and Atmospheric Sciences, Stony Brook University, Stony Brook, NY 11794-5000, USA

Corresponding author: Christopher L. P. Wolfe (Christopher.Wolfe@stonybrook.edu)

Key Points:

- Gulf Stream transport variability decorrelates rapidly with distance downstream of Cape Hatteras.
- Gulf Stream transport downstream of Cape Hatteras is not correlated with sea level on the U.S. East Coast.
- Correlations between the Gulf Stream transport and sea level at Middle Atlantic Bight are results of their concurrent responses to wind.

Abstract

The relationship between Gulf Stream (GS) transport and coastal sea level is investigated using monthly GS transport between 1993–2019 at Florida Straits and ten altimeter tracks. The results show that GS transport decorrelates quickly along its path, indicating it is misleading to assume that transport at a particular location represents strength of the GS as a whole. GS transport south of Cape Hatteras is significantly correlated with coastal sea level in both the South Atlantic Bight (SAB) and Middle Atlantic Bight (MAB). However, the significant correlations in MAB are due to their concurrent response wind on the shelf—the correlation becomes insignificant once the influence of local winds is removed. North of Cape Hatteras, the influence of GS transport on sea level is mostly over the deep ocean and rarely on the shelf, indicating that there is no dynamic link between the GS transport and coastal sea level in this region.

Plain-Language Summary

Sea level on the U.S. East Coast north of Cape Hatteras rose rapidly in recent decades. Some previous studies attributed this sea-level rise to a decline in the Gulf Stream (GS) transport next to it. In this study, we investigate the relation between them from in-situ and remote observations. The results show that GS transport changes continuously along its path, suggesting that the GS strength cannot be represented by its transport at a particular location. South of Cape Hatteras, the GS transport can impact neighboring coastal sea levels via oceanic links. However, its influence on coastal sea level is negligible north of Cape Hatteras. The previously reported relationship between the GS transport and coastal sea level is due to their concurrent responses to winds. The above

results imply that changes in GS transport are unlikely to be the direct cause of rapid sea-level rise at the U.S. East Coast north of Cape Hatteras.

Introduction

Sea level rise on the U.S. East Coast, especially in the Middle Atlantic Bight (MAB), has accelerated over the last few decades and at a rate higher than the global ocean (Boon, 2012; Davis and Vinogradova, 2017; Ezer et al., 2013; Kopp, 2013; Park and Sweet, 2015; Sallenger et al., 2012). Records from tide gauges also show large interannual fluctuations in coastal sea level and the rate of sea level rise (Andres et al., 2013; Ezer, 2013; Goddard et al., 2015). The sea level rise has caused an increase in the number of minor tidal flooding days on the East Coast (Ezer and Atkinson, 2014) and may lead to more frequent extreme high-water events in the future (Tebaldi et al., 2012). It is already a real threat to communities near the coast and has attracted much concern.

Some studies have reported that the Gulf Stream (GS) has weakened in the past two decades (Caesar et al., 2018; Ezer, 2015; Ezer et al., 2013; Smeed et al., 2018), and that changes in GS transport have contributed to both the long-term trend and the interannual fluctuations in sea level along the U.S. East Coast. A weakened GS should be accompanied by a reduction in cross-stream sea level drop due to geostrophy—it has been suggested that this mechanism drives coastal sea rise to the west of the GS (Ezer, 2015; Ezer et al., 2013; Goddard et al., 2015; Yin and Goddard, 2013). This mechanism has also been adopted to explain how a decline in the Atlantic meridional overturning circulation (AMOC) may have contributed to sea level rise on the U.S. East Coast based on the assumption that fluctuations in AMOC transport drive downstream fluctuations in GS transport (Ezer, 2015; Ezer et al., 2013; Ezer and Atkinson, 2014; Goddard et al., 2015; Hu and Bates, 2018; Yin et al., 2010, 2009).

However, Rossby et al. (2014) and Chi et al. (2021) found no significant trend in GS transport from in-situ measurements and altimetry records, respectively, and the lack of a long-term trend makes changes in GS transport a poor explanation for a long-term trend in sea level. Dong et al. (2019) examined the GS transport from altimetry records and also found no link with coastal sea level. Other studies have suggested that local wind plays an important role in sea level fluctuations and correlations between the AMOC, GS and the coastal sea level are due to their concurrent response to large-scale atmospheric circulations (Andres et al., 2013; Piecuch et al., 2019; Valle-Levinson et al., 2017; Woodworth et al., 2014). Little et al. (2019) reviewed the relationship between coastal sea level at the U.S. East Coast, GS, and AMOC in recent studies, pointing out that "the causal relationships between different observational metrics, AMOC, and sea level are often unclear", even though robust correlations can be found.

In this study, we revisit direct links between GS transport and dynamic sea level

at U.S. East Coast. We first derived the GS transport at Florida Straits and 10 satellite altimeter tracks from the Florida Straits to the south of Newfoundland Island. It is found that GS transport does not vary consistently along its path. Then, the relationship between coastal sea level and GS transport at different locations is investigated. We find no significant relationship between GS transport and coastal sea level north of Cape Hatteras once their simultaneous response to wind forcing is accounted for.

Data and Methods

The GS transport through Florida Straits, also known as the Florida Current transport (FCT), has been measured by underwater cables for decades (Meinen et al., 2010). Gaps in the cable records are filled by another dataset derived from sea level differences across the strait (Volkov et al., 2020).

The GS transport downstream of Florida Straits is derived from absolute dynamic topography (ADT) at 10 descending tracks that are approximately perpendicular to the GS path (Figure 1). At each track, cross-track geostrophic velocity is derived from ADT. Then, GS transport is calculated by integrating the geostrophic velocity between the first point where it drops to zero north and south of the GS axis. The GS transport from altimetry is given in units of Sv km⁻¹ (= $10^3 m^2 s^{-1}$) and is proportional to the sea level difference between the two zero velocity points. A decrease of 1 Sv km⁻¹ in GS transport corresponds to a decrease of approximately 0.9 cm in the ADT drop across the GS. The GS transport used in this study is identical to that used in Chi et al. (2021) but extended to 1993–2019. It has been shown that the GS transport from along-track ADT is comparable to in-situ ADCP measurements at the Oleander transect (Chi et al., 2021).

Monthly mean coastal sea level from tide gauges are extracted from the Permanent Service for Mean Sea Level (PSMSL) (Holgate et al., 2013; PSMSL, 2019). Details of the tide gauges are listed in Table S1 and their locations are shown in Figure 1. In this study, we focus on dynamic sea level, which is "the local height of the sea surface above the geoid with the inverse barometer (IB) correction applied" (Gregory et al., 2019). The IB effect is removed using monthly mean sea level pressure from ERA5 (Hersbach et al., 2019), which also provides surface wind stress used in this study.

Results in this study are based on monthly mean data. The annual cycle is removed from both sea level and GS transport by subtracting climate monthly means from the time series. Any trends in sea level and GS transport were not removed in the calculations presented in this paper. We repeated the calculations after removing linear trends, and the results were substantially the same. Statistical significance of correlations is estimated by the random-phase method described in Ebisuzaki (1997), in which data are resampled 20,000 times. The 95% confidence interval is adopted to decide whether a correlation is significant.

Result and discussion

3.1 Streamwise correlations of GS transport

Figure 2 shows correlations between monthly GS transport at different locations. GS transports at different locations are significantly correlated upstream of Cape Hatteras; however, the correlation coefficients are less than 0.5, so less than 25% of the GS transport variance at a given track can be explained by the transport at a neighboring track. Downstream of Cape Hatteras, most correlations of GS transport between neighboring tracks are insignificant. These results show that GS transport varies independently at different locations along its path and the monthly mean transport at one transect is not representative of the monthly mean transport at other transects. In particular, the GS transport upstream of Cape Hatteras, where the GS is closest to the MAB and Gulf of Maine (GoM). This is consistent with the result reported by Sanchez-Franks et al. (2014) that GS transport at Florida Straits and the Oleander transect are not correlated with each other.

With more water joining the GS from recirculation gyres north and south to it, the magnitude of its transport increases dramatically from ~32 Sv at Florida Straits (Meinen et al., 2010) to 85–102 Sv at the Oleander transect (Sanchez-Franks et al., 2014). Thus, recirculation gyre variability likely has a larger impact on GS transport variability downstream of Cape Hatteras than variations in upstream transport. Thus, it is incorrect to attribute a specific fluctuation in transport at a particular location to the GS as a whole. As noted by Stommel (1958), "the GS is not a river of hot water flowing through the ocean." It is a highly turbulent boundary current buffeted by continuously varying topography and instabilities such that its flow decorrelates rapidly as it moves downstream.

3.2 Coastal sea level and GS transport at Florida Straits

Records from both tide gauges and altimetry show that FCT is correlated with sea level along the U.S. East Coast from Florida to Massachusetts (Figure 3a). The significant negative correlations are limited to west of the GS upstream of Cape Hatteras and on the shelf (marked by 1000-meter isobath) in the MAB. This result appears to be consistent with previous studies arguing that the sea level rise in the MAB is partly due to a decrease in GS transport (Ezer et al., 2013).

However, a significant correlation does not necessarily indicate a causal relationship. To further investigate whether the coastal sea level in the MAB and FCT are linked by ocean dynamics or atmospheric circulation, we follow Piecuch et al. (2019) and decompose sea level into a local wind-driven component and a residual component, $h_{\rm res}$. First, the wind direction explaining the largest fraction of local sea level variance is determined by linearly regressing the local wind stress vector against the sea level; then, sea level is regressed against wind stress in that direction. Figure S1 shows that the sea level at all the tide

gauges is significantly correlated with the local wind stress. The direction of greatest correlation is approximately alongshore at most tide gauges from South Atlantic Bight (SAB) to GoM (Figure S1). Sandstrom (1980) suggested that the Coriolis term in alongshore current forced by alongshore wind (and bottom friction) should be balanced by a transverse sea level drop, thus the coastal sea level at the East Coast should be positively correlated with the southwestward alongshore wind. Piecuch et al. (2019) also found that alongshore wind drives local sea level variability on the New England Coast from New York to the GoM, while the influence from cross-shore wind is negligible.

We find that after removing effects of local wind, FCT is significantly correlated with residual sea level at tide gauges only in the SAB (Figure 4a). Thus, the significant correlations between FCT and sea level in the MAB result from their concurrent response to wind. The significant correlation from altimetry next to the coastline around New York City may not be reliable since none of the three tide gauges in the same region show significant correlations.

3.3 Coastal sea level and GS transport derived from along-track ADT

According to geostrophic balance, GS surface transport is proportional to the sea level drop across it; however, significant correlations between transport and sea level do not necessarily extend to the coast. Upstream of Cape Hatteras, the relationship between coastal sea level and GS transport is similar to its relationship with FCT. GS transport at tracks 254 & 76 is negatively correlated with coastal sea level in the SAB and part of the MAB (Figure 3b & Figure S2a).¹ The correlations with tide gauges in the MAB become insignificant once the local wind effects are removed (Figure 4b & Figure S3a).

North of Cape Hatteras, the significant correlations between GS transport and sea level are limited to the deep ocean and rarely appear on the shelf (Figure 3 & Figure S2). The only exception is track 202, where GS transport is correlated with sea level along the slope off MAB after local wind effects are removed (Figure 4d). This may be due to its distinct location—track 202 is located between the Northwest Recirculation Gyre (NWRG, Andres et al., 2020; New et al., 2021) and the Northern Recirculation Gyre (Hogg, 1992; Hogg et al., 1986) (Figure 1). These recirculation gyres allow signals from GS transport at Track 202 to reach the slope more easily than at other locations. This effect can be seen in Figure S4, which shows that sea level changes associated with GS transport extend further north at track 202 than other tracks. GS transport at track 202 is also correlated with sea level near Cape Cod (Figure 4d). However, the small and insignificant correlations between the GS transport and sea level at the three tide gauges near Cape Cod suggest that its influence on coastal sea level is negligible.

With local wind effects removed, GS transport downstream of Cape Hatteras

¹Only results for FCT and GS transport at every other track from altimetry are presented in Figure 3 to keep each panel large enough to be read easily. Results from the remaining tracks are given in the Supporting Information.

is not systemically correlated with sea level at the tide gauges. At the 95% confidence level adopted here, one in twenty correlations are expected to be significant by random chance. With 20 tide gauges shown in Figure 4 and Figure S3, we can expect one significant correlation for each track even if no true correlations exist. We therefore conclude that GS transport downstream of Cape Hatteras is not a major driver for coastal sea level north of it.

The above conclusion is reasonable since a large part of the GS transport downstream of Cape Hatteras is from recirculation gyres instead of basin scale circulations described in classic theories. It is the sea level at centers of the recirculation gyres, which is also the edge of GS, that matters for GS transport. Sea level variations at centers of the recirculation gyres are not necessarily related to sea level variations at their edges, or at the coastline. Composites of sea level profiles across the GS (Figure S4) show that the sea level rise (decrease) north of the GS associated with decreased (increased) GS transport decays away from the GS and becomes negligible approximately 300 km north of the GS axis at all tracks downstream of Cape Hatteras. The above result is also consistent with the correlation map shown in Figure S5. The significant positive correlations between mean sea level from tide gauges at MAB (GoM) and gridded sea level from altimetry are restricted to the shelf.

Overall, the GS transport only affects coastal sea level south of Cape Hatteras. North of Cape Hatteras, its influence on sea level is mostly over the deep ocean and does not reach the coastline.

Summary

The Gulf Stream (GS) transport during 1993–2019 from underwater cables at Florida Straits and satellite altimeters at 10 descending tracks from Florida to Canada coast are investigated in this study. Only a few statistically significant correlations between GS transport at different locations are found. Correlations, even where significant, are less than 0.5, implying that more than 75 percent of variations of GS transport, even between neighboring locations, are independent of each other. Hence, it is misleading to suggest that transport variations at any particular transect represent the GS as a whole.

The relationship between GS transport along its path and sea level at the U.S. East Coast during 1993–2019 was examined. South of Cape Hatteras, coastal sea level is significantly anti-correlated with adjacent GS transport, indicating that a decrease in GS transport can lead to sea-level rise to its west via geostrophy. This mechanism has been adopted in some previous studies (Ezer, 2013; Ezer et al., 2013; Yin et al., 2009) to suggest that a decrease in GS transport will affect sea levels along the U.S. East Coast. However, north of Cape Hatteras—which is a "hotspot" of sea level rise in recent decades—the GS's transport influence on sea level is restricted to the deep ocean. The coastal sea level in this region is rarely correlated with GS transport, which indicates that

the above mechanism is not generally applicable. Even though significant correlation can be found between the GS transport near Florida Straits and sea level in Middle Atlantic Bight (MAB), such correlations become insignificantly small when local wind effects are removed, indicating the significant correlations are due to their concurrent responses to wind rather than a direct dynamic link.

Some previous studies suggested that a decrease in GS transport, accompanied by a weakened AMOC, contributed to the rapid sea level rise in MAB and further north (Ezer, 2015, 2013, 2001; Ezer et al., 2013; Goddard et al., 2015; Yin et al., 2009; Yin and Goddard, 2013). However, the lack of direct link between the GS transport and coastal sea level north of Cape Hatteras presented in this study indicates that this is unlikely to be true. By analyzing 20-year ADCP measurements at the Oleander transect, Rossby et al. (2014) found that there is no long-term trend in the GS transport. This is also confirmed by two follow-up investigations (Chi et al., 2021; Dong et al., 2019), both of which show stable GS transport west of 70°W. Thus, it is not likely that changes in GS transport contribute to the sea level rise at U.S. East Coast. The horizontal resolution of most numerical models in previous studies is 1° or coarser. Chi et al. (2018) showed that such coarse resolution models cannot reproduce the GS structure and recirculation gyres north of the GS correctly. Also, the shelf off the U.S. East Coast is about 100~200 km in width, which is only $1\sim2$ grid points in the models. Thus, the models are not expected to simulate the processes on the shelf accurately.

Even though significant correlations between coastal sea level and the GS transport are rarely found north of Cape Hatteras, the GS may still affect the coastal sea level indirectly. Changes in heat transported by the GS may drive steric sea level changes in the northwest Atlantic. The GS may also affect coastal sea level indirectly via its interaction with the Labrador Current near Grand Banks, since Frederikse et al. (2017) and Gonçalves Neto et al. (2021) suggest that the Labrador Current might play a role in sea level variations on the shelf. Wise et al. (2020) also indicated that sea level variability north of Cape Hatteras is driven by the subpolar gyre. Limited by the available data, only the geostrophic component of GS transport is discussed in this study. Little et al. (2019) suggested that the ageostrophic component might be important and worth further investigation.

Acknowledgement

This work was supported by the NSF (OCE-1634829). The along-track absolute dynamic topography is extracted from *Global Ocean Along-track L3 Sea Surface Heights Reprocessed (1993-Ongoing) Tailored for Data Assimilation* (https://resources.marine.copernicus.eu/product-detail/SEALEVEL_GLO_PHY_L 3_REP_OBSERVATIONS_008_062/INFORMATION), and the gridded absolute dynamic topography is extracted from *Global Ocean Gridded L4 Sea Surface Heights and Derived Variables Reprocessed (1993-Ongoing)* (https://re sources.marine.copernicus.eu/product-detail/SEALEVEL_GLO_PHY_L4_R EP_OBSERVATIONS_008_047/INFORMATION). The Florida Current

transport from cable measurements is available from the Atlantic oceanographic and Meteorological Laboratory (https://www.aoml.noaa.gov/phod/floridacu rrent/data_access.php). The sea level pressure and surface wind stress are extracted from *ERA5 Monthly Averaged Data on Single Levels from 1979 to Present* via Copernicus Climate Change Service Climate Data Store (https: //cds.climate.copernicus.eu/cdsapp#!/dataset/reanalysis-era5-single-levelsmonthly-means?tab=overview).

References

Andres, M., Gawarkiewicz, G.G., Toole, J.M., 2013. Interannual sea level variability in the western North Atlantic: Regional forcing and remote response. Geophys. Res. Lett. 40, 5915–5919. https://doi.org/10.1002/2013GL058013

Boon, J.D., 2012. Evidence of Sea Level Acceleration at U.S. and Canadian Tide Stations, Atlantic Coast, North America. J. Coast. Res. 28, 1437–1445. https://doi.org/10.2112/JCOASTRES-D-12-00102.1

Caesar, L., Rahmstorf, S., Robinson, A., Feulner, G., Saba, V., 2018. Observed fingerprint of a weakening Atlantic Ocean overturning circulation. Nature 556, 191–196. https://doi.org/10.1038/s41586-018-0006-5

Chi, L., Wolfe, C.L.P., Hameed, S., 2021. Has the Gulf Stream Slowed or Shifted in the Altimetry Era? Geophys. Res. Lett. 48, e2021GL093113. https://doi.org/10.1029/2021GL093113

Chi, L., Wolfe, C.L.P., Hameed, S., 2018. Intercomparison of the Gulf Stream in ocean reanalyses: 1993-2010. Ocean Model. 125, 1-21. https://doi.org/10.1016/j.ocemod.2018.02.008

Davis, J.L., Vinogradova, N.T., 2017. Causes of accelerating sea level on the East Coast of North America. Geophys. Res. Lett. 44, 5133–5141. https://doi.org/10.1002/2017GL072845

Dong, S., Baringer, M.O., Goni, G.J., 2019. Slow down of the Gulf Stream during 1993–2016. Sci. Rep. 9, 6672. https://doi.org/10.1038/s41598-019-42820-8

Ebisuzaki, W., 1997. A method to estimate the statistical significance of a correlation when the data are serially correlated. J. Clim. 10, 2147–2153. https://doi.org/10.1175/1520-0442(1997)010<2147:AMTETS>2.0.CO;2

Ezer, T., 2015. Detecting changes in the transport of the Gulf Stream and the Atlantic overturning circulation from coastal sea level data: The extreme decline in 2009–2010 and estimated variations for 1935–2012. Glob. Planet. Change 129, 23–36. https://doi.org/10.1016/j.gloplacha.2015.03.002

Ezer, T., 2013. Sea level rise, spatially uneven and temporally unsteady: Why the U.S. East Coast, the global tide gauge record, and the global altimeter data show different trends: SPATIAL AND TEMPORAL SEA LEVEL RISE. Geophys. Res. Lett. 40, 5439–5444. https://doi.org/10.1002/2013GL057952

Ezer, T., 2001. Can long-term variability in the Gulf Stream Transport be inferred from sea level? Geophys. Res. Lett. 28, 1031–1034. https://doi.org/10.1029/2000GL011640

Ezer, T., Atkinson, L.P., 2014. Accelerated flooding along the U.S. East Coast: On the impact of sea-level rise, tides, storms, the Gulf Stream, and the North Atlantic Oscillations. Earths Future 2, 362–382. https://doi.org/10.1002/2014EF000252

Ezer, T., Atkinson, L.P., Corlett, W.B., Blanco, J.L., 2013. Gulf Stream's induced sea level rise and variability along the U.S. mid-Atlantic coast. J. Geophys. Res. Oceans 118, 685–697. https://doi.org/10.1002/jgrc.20091

Frederikse, T., Simon, K., Katsman, C.A., Riva, R., 2017. The sea-level budget along the Northwest Atlantic coast: GIA, mass changes, and large-scale ocean dynamics. J. Geophys. Res. Oceans 122, 5486–5501. https://doi.org/10.1002/2017JC012699

Goddard, P.B., Yin, J., Griffies, S.M., Zhang, S., 2015. An extreme event of sea-level rise along the Northeast Coast of North America in 2009–2010. Nat. Commun. 6, 6346. https://doi.org/10.1038/ncomms7346

Gonçalves Neto, A., Langan, J.A., Palter, J.B., 2021. Changes in the Gulf Stream preceded rapid warming of the Northwest Atlantic Shelf. Commun. Earth Environ. 2, 1–10. https://doi.org/10.1038/s43247-021-00143-5

Gregory, J.M., Griffies, S.M., Hughes, C.W., Lowe, J.A., Church, J.A., Fukimori, I., Gomez, N., Kopp, R.E., Landerer, F., Cozannet, G.L., Ponte, R.M., Stammer, D., Tamisiea, M.E., van de Wal, R.S.W., 2019. Concepts and Terminology for Sea Level: Mean, Variability and Change, Both Local and Global. Surv. Geophys. 40, 1251–1289. https://doi.org/10.1007/s10712-019-09525-z

Hersbach, H., Bell, B., Berrisford, P., Biavati, G., Horányi, A., Muñoz Sabater, J., Nicolas, J., Peubey, C., Radu, R., Rozum, I., Schepers, D., Simmons, A., Soci, C., Dee, D., Thépaut, J.-N., 2019. ERA5 monthly averaged data on single levels from 1979 to present, Copernicus Climate Change Service (C3S) Climate Data Store (CDS). (Accessed on 27-Aug-2021), https://doi.org/10.24381/cds.f17050d7

Hogg, N.G., 1992. On the transport of the Gulf Stream between Cape Hatteras and the Grand Banks. Deep Sea Res. Part Oceanogr. Res. Pap. 39, 1231–1246. https://doi.org/10.1016/0198-0149(92)90066-3

Hogg, N.G., Pickart, R.S., Hendry, R.M., Smethie, W.J., 1986. The northern recirculation gyre of the gulf Stream. Deep Sea Res. Part Oceanogr. Res. Pap. 33, 1139–1165. https://doi.org/10.1016/0198-0149(86)90017-8

Holgate, S.J., Matthews, A., Woodworth, P.L., Rickards, L.J., Tamisiea, M.E., Bradshaw, E., Foden, P.R., Gordon, K.M., Jevrejeva, S., Pugh, J., 2013. New data systems and products at the permanent service for mean sea level. J. Coast. Res. 29, 493–504. https://doi.org/10.2112/JCOASTRES-D-12-00175.1

Hu, A., Bates, S.C., 2018. Internal climate variability and projected future regional steric and dynamic sea level rise. Nat. Commun. 9, 1068. https://doi.org/10.1038/s41467-018-03474-8

Kopp, R.E., 2013. Does the mid-Atlantic United States sea level acceleration hot spot reflect ocean dynamic variability? Geophys. Res. Lett. 40, 3981–3985. https://doi.org/10.1002/grl.50781

Little, C.M., Hu, A., Hughes, C.W., McCarthy, G.D., Piecuch, C.G., Ponte, R.M., Thomas, M.D., 2019. The relationship between U.S. East Coast sea level and the Atlantic Meridional Overturning Circulation: A review. J. Geophys. Res. Oceans 2019JC015152. https://doi.org/10.1029/2019JC015152

Meinen, C.S., Baringer, M.O., Garcia, R.F., 2010. Florida Current transport variability: An analysis of annual and longer-period signals. Deep Sea Res. Part Oceanogr. Res. Pap. 57, 835–846. https://doi.org/10.1016/j.dsr.2010.04.001

Park, J., Sweet, W., 2015. Accelerated sea level rise and Florida Current transport. Ocean Sci. 11, 607–615. https://doi.org/10.5194/os-11-607-2015

Piecuch, C.G., Dangendorf, S., Gawarkiewicz, G.G., Little, C.M., Ponte, R.M., Yang, J., 2019. How is New England coastal sea level related to the Atlantic Meridional Overturning Circulation at 26°N? Geophys. Res. Lett. 46, 5351–5360. https://doi.org/10.1029/2019GL083073

Rossby, T., Flagg, C.N., Donohue, K., Sanchez-Franks, A., Lillibridge, J., 2014. On the long-term stability of Gulf Stream transport based on 20 years of direct measurements. Geophys. Res. Lett. 41, 114–120. https://doi.org/10.1002/2013GL058636

Sallenger, A.H., Doran, K.S., Howd, P.A., 2012. Hotspot of accelerated sea-level rise on the Atlantic coast of North America. Nat. Clim. Change 2, 884–888. https://doi.org/10.1038/nclimate1597

Sanchez-Franks, A., Flagg, C.N., Rossby, T., 2014. A comparison of transport and position between the Gulf Stream east of Cape Hatteras and the Florida Current. J. Mar. Res. 72, 291–306. https://doi.org/10.1357/002224014815460641

Sandstrom, H., 1980. On the wind-induced sea level changes on the Scotian shelf. J. Geophys. Res. Oceans 85, 461–468. https://doi.org/10.1029/JC085iC01p00461

Smeed, D.A., Josey, S.A., Beaulieu, C., Johns, W.E., Moat, B.I., Frajka-Williams, E., Rayner, D., Meinen, C.S., Baringer, M.O., Bryden, H.L., Mc-Carthy, G.D., 2018. The North Atlantic Ocean Is in a state of reduced overturning. Geophys. Res. Lett. 45, 1527–1533. https://doi.org/10.1002/2017GL076350

Stommel, H.M., 1958. The Gulf Stream: a physical and dynamical description. University of California Press, Berkeley.

Tebaldi, C., Strauss, B.H., Zervas, C.E., 2012. Modelling sea level rise impacts on storm surges along US coasts. Environ. Res. Lett. 7, 014032. https://doi.org/10.1088/1748-9326/7/1/014032 Valle-Levinson, A., Dutton, A., Martin, J.B., 2017. Spatial and temporal variability of sea level rise hot spots over the eastern United States. Geophys. Res. Lett. 44, 7876–7882. https://doi.org/10.1002/2017GL073926

Volkov, D.L., Domingues, R., Meinen, C.S., Garcia, R., Baringer, M., Goni, G., Smith, R.H., 2020. Inferring Florida Current Volume Transport From Satellite Altimetry. J. Geophys. Res. Oceans 125, e2020JC016763. https://doi.org/10.1029/2020JC016763

Wise, A., Polton, J.A., Hughes, C.W., Huthnance, J.M., 2020. Idealised modelling of offshore-forced sea level hot spots and boundary waves along the North American East Coast. Ocean Model. 155, 101706. https://doi.org/10.1016/j.ocemod.2020.101706

Woodworth, P.L., Maqueda, M.Á.M., Roussenov, V.M., Williams, R.G., Hughes, C.W., 2014. Mean sea-level variability along the northeast American Atlantic coast and the roles of the wind and the overturning circulation. J. Geophys. Res. Oceans 119, 8916–8935. https://doi.org/10.1002/2014JC010520

Yin, J., Goddard, P.B., 2013. Oceanic control of sea level rise patterns along the East Coast of the United States. Geophys. Res. Lett. 40, 5514–5520. https://doi.org/10.1002/2013GL057992

Yin, J., Griffies, S.M., Stouffer, R.J., 2010. Spatial variability of sea level rise in twenty-first century projections. J. Clim. 23, 4585–4607. https://doi.org/10.1175/2010JCLI3533.1

Yin, J., Schlesinger, M.E., Stouffer, R.J., 2009. Model projections of rapid sealevel rise on the northeast coast of the United States. Nat. Geosci. 2, 262–266. https://doi.org/10.1038/ngeo462



Figure 1: (Background shading) Topography of the northwest Atlantic. The solid (dashed) purple lines are contours for positive (negative) absolute dynamic topography (ADT) with a 10-cm interval during 1993-2019. The mean GS path is marked by the thick red line. Satellite tracks are marked by white lines and tide gauges used in this study are marked by cyan dots. The Track 152 is not used in this study since the northern edge of the GS is too close to the coastline.



Figure 2: Correlation coefficients between monthly mean GS transport at locations. The correlations significant at 95% level are marked by black dots. Florida Current Transport (FCT) represents the GS transport at the Florida Straits.



Figure 3: Correlation between monthly averaged ADT and (a) GS transport at Florida Straits, (b-f) surface-layer Gulf Stream transport at altimetry tracks. The correlations significant at the 95% confidence level are bounded by black contours. Locations of tide gauges are marked by circles and correlations between sea level from tide gauges and the Gulf Stream transport are indicated by the color inside those circles. Orange (blue) triangles (circles) indicate statistically significant (insignificant) correlations. The violet line indicates the mean GS path. The dashed black line indicates the altimetry track. The grey solid line indicates 1000-meter isobath, which is approximately the edge of the

shelf. The plots have been rotated to show the entire U.S. East Coast. Some major landmarks are noted, among which CH indicates Cape Hatteras and NY indicates New York City.



Figure 4: Same as Figure 3 but for residual sea level with local wind effects removed.